

Citation: Rasmussen ZA, Shah WH, Hansen CL, Azam SI, Hussain E, Schaefer BA, et al. (2021) Examining the relationships between early childhood experiences and adolescent and young adult health status in a resource-limited population: A cohort study. PLoS Med 18(9): e1003745. https://doi.org/10.1371/journal.pmed.1003745

Academic Editor: Kathryn Mary Yount, Emory University, UNITED STATES

Received: February 23, 2021

Accepted: July 28, 2021

Published: September 28, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003745

Copyright: © 2021 Rasmussen et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data is held in a public repository entitled the National Institute of Child Health and Human Development Data and

RESEARCH ARTICLE

Examining the relationships between early childhood experiences and adolescent and young adult health status in a resourcelimited population: A cohort study

Zeba A. Rasmussen^{1*}, Wasiat H. Shah^{2°}, Chelsea L. Hansen^{1°}, Syed Iqbal Azam^{3°}, Ejaz Hussain⁴, Barbara A. Schaefer⁵, Nicole Zhong^{5¤a}, Alexandra F. Jamison¹, Khalil Ahmed⁶, Benjamin J. J. McCormick¹, for the Oshikhandass Water, Sanitation, Health and Hygiene Interventions Project

1 Division of International Epidemiology and Population Studies, Fogarty International Center, National Institutes of Health, Bethesda, Maryland, United States of America, 2 Department of Medicine, Aga Khan University, Karachi, Pakistan, 3 Department of Community Health Sciences, Aga Khan University, Karachi, Pakistan, 4 Administration Department, Karakoram International University, Gilgit, Pakistan, 5 Department of Educational Psychology, Counseling, and Special Education, Pennsylvania State University, University Park, Pennsylvania, United States of America, 6 Faculty of Life Sciences, Karakoram International University, Gilgit, Pakistan

• These authors contributed equally to this work.

¤a Current address: Department of Human Services, School of Education and Human Development, University of Virginia, Charlottesville, Virginia, United States of America

* zeba.rasmussen@nih.gov

Abstract

Background

Adolescence is a critical point in the realization of human capital, as health and educational decisions with long-term impacts are made. We examined the role of early childhood experiences on health, cognitive abilities, and educational outcomes of adolescents followed up from a longitudinal cohort study in Pakistan, hypothesizing that early childhood experiences reflecting poverty would manifest in reduced health and development in adolescence.

Methods and findings

Adolescents/young adults previously followed as children aged under 5 years were interviewed. Childhood data were available on diarrhea, pneumonia, and parental/household characteristics. New data were collected on health, anthropometry, education, employment, and languages spoken; nonverbal reasoning was assessed. A multivariable Bayesian network was constructed to explore structural relationships between variables. Of 1,868 children originally enrolled, 1,463 (78.3%) were interviewed as adolescents (range 16.0–29.3 years, mean age 22.6 years); 945 (65%) lived in Oshikhandass. While 1,031 (70.5%) of their mothers and 440 (30.1%) of their fathers had received no formal education, adolescents reported a mean of 11.1 years of education. Childhood diarrhea (calculated as episodes/child-year) had no association with nonverbal reasoning score (an arc was supported in just 4.6% of bootstrap samples), health measures (with BMI, 1% of bootstrap samples; Specimen Hub (DASH) at https://dash.nichd.nih. gov/study/416387.

Funding: The study was funded by the Pakistan US S&T Cooperative Agreement (https://sites. nationalacademies.org/pga/pakistan/index.htm) between the Pakistan Higher Education Commission (HEC, https://www.hec.gov.pk/ english/pages/home.aspx) (No.4-421/PAK-US/ HEC/2010/955, grant to the Karakoram International University, to KA) and US National Academies of Science (https://www. nationalacademies.org/) (Grant Number PGA-P211012 from NAS to the Fogarty International Center, to ZAR). Original study funding: The Applied Diarrheal Disease Research Program at Harvard Institute for International Development (Grants 063 and P033 to ZAR), and the Aga Khan Health Service, Northern Areas and Chitral, Pakistan (https://www.akdn.org/aga-khan-healthservice-pakistan-0, in kind support to ZAR). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: BMI, body mass index; BN, Bayesian network; G-B, Gilgit-Baltistan; HAZ, length/height-for-age *z* score; HCI, Human Capital Index; SRPH, self-reported past childhood health; SRCH, self-reported current health. systolic and diastolic blood pressure, 0.1% and 1.6% of bootstrap samples, respectively), education (0.7% of bootstrap samples), or employment (0% of bootstrap samples). Relationships were found between nonverbal reasoning and adolescent height (arc supported in 63% of bootstrap samples), age (84%), educational attainment (100%), and speaking English (100%); speaking English was linked to the childhood home environment, mediated through maternal education and primary language. Speaking English (n = 390, 26.7% of adolescents) was associated with education (100% of bootstrap samples), self-reported child health (82%), current location (85%) and variables describing childhood socioeconomic status. The main limitations of this study were the lack of parental data to characterize the home setting (including parental mental and physical health, and female empowerment) and reliance on self-reporting of health status.

Conclusions

In this population, investments in education, especially for females, are associated with an increase in human capital. Against the backdrop of substantial societal change, with the exception of a small and indirect association between childhood malnutrition and cognitive scores, educational opportunities and cultural language groups have stronger associations with aspects of human capital than childhood morbidity.

Author summary

Why was this study done?

- Childhood diarrhea and pneumonia remain leading causes of morbidity and mortality, and their effects are hypothesized to have long-lasting impacts that could potentially negatively affect adolescent/young adult nutrition, reasoning skills, and educational status.
- Studies across different geographical settings have focused on outcomes during early childhood and on nutritional status or interventions, though very few have extended into adolescence/young adulthood.
- Other studies have noted negative associations between poverty, morbidity, and malnutrition and childhood attainment and physical and cognitive development.

What did the researchers do and find?

- We conducted a prospective longitudinal study on a cohort of children (n = 1,868, 1989–1996) from a poor, remote rural Pakistani village who were followed up 20 years later (n = 1,463, 2011-2014), when they were adolescents and young adults (mean age 22.6 years old).
- Most of their mothers (70.5%) and fathers (30.1%) had received no formal education, but the adolescents had, on average, 11.1 years of education.

- Childhood illness was not related to nonverbal reasoning score, health measures, education, or employment.
- Education (including speaking English) was strongly related to nonverbal reasoning score.
- Speaking English was related to improved childhood socioeconomic status, better child health, more years of education by adolescence, and greater likelihood of having left the village.

What do these findings mean?

- Investment in education, especially for females, is important in the development of human capital.
- Childhood undernutrition was related to adolescent height, and thereby weakly to cognitive score, whereas educational opportunities and cultural language group were more strongly associated with human capital than childhood illness was.
- This population underwent substantial secular changes, including economic growth and expanded educational opportunities, that may have overcome deficits from child-hood undernutrition.

Introduction

Human capital is the knowledge, skills, and health that individuals accumulate over time that enables them to realize their potential as productive members of society [1]. The World Bank created in 2018 a population-level indicator of human capital by indexing child survival and stunting, school attendance and performance, and adult survival. Although essentially an economic construct, its goal is to capture shortfalls in the realization of human development. The Human Capital Index (HCI) describes how improvements to childhood health, nutrition, and early learning lay strong foundations for the future acquisition of cognitive, social, and behavioral skills [2]. As such, the adolescent period is a critical stage between childhood and adulthood where formal education ends and key health decisions are made that have long-term impacts [3,4].

Pakistan ranks 134 of 157 countries in the 2018 World Bank HCI [1]. Located in remote, mountainous northeast Pakistan, Gilgit-Baltistan (G-B) in the 1980s was one of the poorest regions of the country, with a per capita income less than half that of the rest of Pakistan [5], with subsistence farming as the primary occupation, low educational levels, low life expectancy (53 years), and high maternal mortality [6]. Considerable investment in rural support, education, and health changed this situation after the 1978 opening of the Karakoram Highway, which provided increased accessibility to the area, leading to substantial societal change [5]. Communities in G-B were strongly motivated to invest in children's education for future job opportunities and income [7]. The most recent evaluation in 2010 showed a dramatic secular transformation, resulting in a narrowing of the income gap, to a per capita income of about 90% of the national average [8].

We report the findings of an observational longitudinal study of a cohort of children who were followed up 15–20 years later, when they were adolescents and young adults (hereafter referred to as adolescents) [4]. The childhood cohort came from Oshikhandass, an ethnically mixed, remote rural village with high morbidity and mortality [9], 20 km from the capital of G-B. This study examines individual-level associations between childhood health and family socioeconomic indicators of early life adversity and adolescent health and development. Our study considers the following as outcomes of interest: nonverbal reasoning score, self-reported health status, body mass index (BMI), blood pressure, educational attainment, and employment. We hypothesized that early life experiences reflecting poverty would manifest in reduced health and development in adolescence, which are the foundation of adult human capital. [10].

Methods

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guidelines (see <u>S1 Appendix</u>) [11]. Prospective protocols (see S2 and S5 Appendices) were used for the collection of data but not for analysis.

Ethics approval and consent to participate

Ethical approval was granted for the 1989–1996 study by the Aga Khan University Human Subjects Protection Committee (15 November 1989 for childhood diarrhea; 3 November 1993 for pneumonia); parents/legal guardians of children provided oral informed consent to participate, given high levels of parental illiteracy. Ethical approval for the adolescent study was granted by the US National Institutes of Health Institutional Review Board (#20TWN071, initially approved by the National Institute of Child Health and Development's institutional review board), the Aga Khan University Ethics Review Committee (1966-CHS-ERC-11), and Karakoram International University Ethics Review Committee. For the adolescent study, all participants aged 18 years or over provided signed consent to participate; participants under age 18 years provided assent, and their parents provided written consent for their participation.

Data collection for children under 5 years, 1989-1996

From 1989 to 1996, all children under age 5 years in Oshikhandass were enrolled with parental consent in the prospective cohort of the Oshikhandass Diarrhea and Pneumonia Project. Local health workers visited families weekly until the child's fifth birthday, out-migration, or death; data collection methods and variable definitions are described elsewhere [9]. These data were used to estimate the burden of child morbidity (number of diarrhea and pneumonia episodes divided by days of follow-up). Pneumonia was not included in the main analyses because of a smaller sample size [9]. The mean length/height-for-age *z* score (HAZ) was calculated for each child from their available quarterly data, using 2006 WHO growth standards. At enrollment, a survey was conducted of household characteristics (e.g., house type based on construction materials, number of rooms and occupants, and type of water and sanitation) and parental educational attainment (illiterate/no formal education, primary to 10 years, or more), occupation, and income [9].

Data collection for follow-up in adolescence and young adulthood, 2011–2014

In July 2011, a village census identified participants from the original cohort who still lived in Oshikhandass, had migrated elsewhere, or died. Cohort members were contacted through their families. After obtaining written informed consent from adolescents, interviews were

conducted in Urdu and, rarely, in the participant's mother tongue (usually Burushaski or Shina), by study staff in person at the study office, at the participant's residence, or by telephone (<1%).

A questionnaire (see <u>S3</u> Appendix) was used to collect data on current and past health status. The questionnaire was specifically designed for this study, considering important domains identified in the literature and in consultation with local subject specialists. The questionnaire was translated into Urdu, back-translated, and piloted in a neighboring village. Self-reported current health (SRCH) and self-reported past childhood health (SRPH) (between ages 5 and 15 years) were characterized using a 5-point scale (very poor to excellent), and other lifetime health problems were recorded as "any" versus "none." The questionnaire recorded educational attainment (years of schooling), whether classes were repeated, employment (student, employed, or not employed), marital status, number of children, current geographic location, age when moved, and languages spoken. Primary language (i.e., mother tongue) was defined as Burushaski or Shina; almost all respondents spoke Urdu, the national language.

The Raven's Standard Progressive Matrices and Colored Progressive Matrices (PsychCorp) [12] were administered (see <u>S5 Appendix</u>) in a quiet location without distractions to assess nonverbal reasoning skills, which are a critical component of cognitive ability [13]. The Raven's matrices are widely used internationally because they have minimal language requirements, thus minimizing the impact of cultural and linguistic differences [12]. Scores were calculated by adding the number of correct items and then scaled into a T score (mean 50, standard deviation 10). Scores were analyzed using exploratory and confirmatory factor analytic approaches to ensure construct validity [14] (see <u>S4 Appendix</u>).

Clinical measurements, taken once each by trained master's level field coordinators, included participant weight (Seca 872 digital scales) without shoes and in light clothing, height (HM200P PortStad Portable Stadiometer, Charder), and abdominal girth (tape measure). BMI was calculated, analyzed as a continuous variable, and for context summarized as underweight (<18.5 kg/m²), normal (18.5–24.9 kg/m²), overweight (25.0–29.9 kg/m²), and obese (\geq 30 kg/m²); pregnant women were excluded. Blood pressure (in mm Hg) was measured once while the rested participant was sitting cross-legged on the floor using a mercury sphygmomanometer (Yamasu Model 600, Japan) and defined as normal except if elevated (systolic blood pressure 121–130 mm Hg and diastolic blood pressure \leq 80 mm Hg), stage 1 hypertension (systolic blood pressure > 140 mm Hg or diastolic blood pressure > 90 mm Hg). Missing data were excluded from the analysis described below; for example, clinical measurements and Raven's matrices were not available for adolescents interviewed by telephone.

Data analysis

Univariate linear models were constructed using the Raven's score (squared to approximate normality) as the outcome. Subsequently, a multivariate Bayesian network (BN, using the R package bnlearn) [15] was constructed to explore the structural relationships between variables; BN analysis determines an optimal statistical description of all the data as it considers many possible model structures [16]. The best-fitting structure to describe the data was sought using a tabu greedy search [17], assuming either linear regression for continuous nodes or conditional probabilities for categorical (including binary) nodes, and a ban list to exclude directed arcs from any variable to the age and sex of the individual (biologically implausible) or from descriptions of their current health, education, or employment status to the variables characterizing their childhood environment (temporally impossible). The single best-fitting model, optimizing the Bayesian information criterion, was bootstrapped (7,500 times), and

only arcs present in 50% or more of samples were retained. To estimate confidence intervals for the primary outcomes, regression models were separately constructed using subnetworks identified from the BN.

All analyses were conducted in R (April 2018, release 3.5.0).

Results

Of 1,868 children enrolled in the initial study, 1,463 (78.3%) were interviewed as adolescents (Fig 1). Almost two-thirds (n = 945) were still living in Oshikhandass, 35.1% (347) elsewhere within Pakistan, and 0.3% (5) abroad. Seventy-nine percent had complete information to build the BN (n = 1,165). Compared to the 1,462 individuals who were re-enrolled as adolescents, individuals lost to follow-up (n = 381) tended to be male (57.2% versus 51.0%, p = 0.017, chisquared test) and to have more diarrheal episodes per child-year (mean [95% CI] 0.54 [0.00, 1.42] versus 0.42 [0.00, 1.02], p = 0.017, Kruskal–Wallis test), less improved household construction (62.2% improved or somewhat improved versus 65.7%, p = 0.002, chi-squared test), and more likely to have improved toilets (3.9% versus 2.3%, p = 0.032, chi-squared test) (see S1 Table). Of 126 (6.8%) children with follow-up who had died, 95 died during the first study and 31 died between studies. Those who did not complete the Raven's matrices (n = 60/1,462, 4.1%, most because they reported a lack of time), compared to those who did complete the Raven's matrices, were more likely to live away from Oshikhandass (63.3% versus 34.2%, p < 0.001, chi-squared test), were older (mean 26.7 versus 22.2 years old, p < 0.001, Kruskal– Wallis test), had a higher BMI (mean 23.0 versus 21.7 kg/m², p = 0.002, *t*-test), had fewer years of education (mean 7.9 versus 11.2 years, p < 0.001, *t*-test), and were less likely to speak English (16.7% versus 27.1%, *p* = 0.10, chi-squared test).

Selected cohort characteristics are shown in Table 1. Males outnumbered females in both the original and adolescent cohorts, despite more male deaths (71 male versus 55 female). Mean age at follow-up was 22.6 years (range 16.0–29.3) based on recorded birthdate. Only 14.4% were married; however, this percentage was higher among the female participants, who were married longer (mean 3.8 versus 2.2 years, p < 0.001, *t*-test), were married at a younger age (mean 21.6 versus 24.7 years, p < 0.001, *t*-test), and had more children (mean 1.8 versus 1.3, p = 0.01, t-test). At the time of interview, 54 women (7.5%) were pregnant. Females were more likely to remain in Oshikhandass, whereas males were more likely to move for education (30.6% versus 9.6%, p < 0.001, test of proportions) or employment (8.4% versus 0.8%, p < 0.001, test of proportions)0.001, test of proportions). In total, 1,240 (67.3%) adolescents had at least 1 childhood HAZ observation (median 8.0, IQR 9), with a population mean HAZ of -2.2, but the mean adolescent HAZ (assuming an age of 19 years, since WHO growth reference standards for calculating age- and sex-standardized HAZ only go up to this age) was -1 (see S1 Fig). Childhood HAZ was positively correlated with adolescent height (Pearson's rho 0.15 for females and 0.24 for males). Along with Urdu, almost all adolescents spoke Shina (1,414, 96.7%), the language most widely spoken in Gilgit, and about two-thirds also spoke Burushaski, the mother tongue of those originally from Hunza. A quarter (367, 25.1%) of adolescents spoke \geq 4 languages, and the fourth most common language was English, spoken more often by males. In our population, based on a linear regression adjusting for age and sex, speaking English was associated with having approximately 2.4 more years of education (95% CI 2.1 to 2.7, p < 0.001); all English speakers had ≥ 8 years of education.

Nonverbal reasoning score using the Raven's matrices

In univariate linear regressions (see S2 Table), the Raven's score (squared T score) was significantly ($p \le 0.05$) negatively related to childhood household density (-160 [95% CI -220, -91],



Fig 1. Flowchart of participant follow-up.

https://doi.org/10.1371/journal.pmed.1003745.g001

p < 0.001), inferior household (-400 [95% CI -530, -260], p < 0.001) or toilet construction (-650 [95% CI -980, -320], p = 0.003), lower status paternal occupation (-420 [95% CI -550, -280], p < 0.001), whether they were a Shina speaker (-440 [95% CI -550, -340], p < 0.001), and reported worse health (SRCH, -440 [95% CI -770, -370], *p* < 0.001; SRPH, -660 [95% CI -850, -460, p < 0.001; scores were positively related to being male (280 [95% CI 180, 370], p < 0.001, height (21 [95% CI 16, 26], p < 0.001), weight (13 [95% CI 7.6, 18], p < 0.001), waist girth (5.2 [95% CI 0.11, 10], p = 0.045), moving and living away from Oshikhandass (730 [95% CI 560, 900], p < 0.001), age at moving (408 [95% CI 249, 567], p < 0.001), whether they spoke English (850 [95% CI 750, 950], *p* < 0.001), being unmarried (350 [95% CI 200, 490], *p* < 0.001), being employed or a student (380 [95% CI 220, 540] and 550 [95% CI 410, 690], respectively, p < 0.001), having more years of education (130 [95% CI 110, 140], p < 0.001), not having repeated any classes (150 [95% CI 55, 250], p = 0.002), parents' educational status (maternal, 870 [95% CI 580, 1,200], *p* < 0.001; paternal, 610 [95% CI 470, 750], *p* < 0.001), maternal occupation (occupation other than housewife, 710 [95% CI 520, 900], p < 0.001), and higher parental income (320 [95% CI 150, 500], p < 0.001). The number of variables related to the Raven's score was substantially reduced in the multivariate BN, with T score directly associated with only an individual's adolescent height, age at assessment, educational attainment, and whether or not they spoke English (Fig 2). Older participants tended to have a lower Raven's score (-41.1 squared T score points per year [95% CI -55.8, -26.4], p < 0.001, linear regression), but taller participants and those with more years of education achieved higher scores (13.3 squared T score points per centimeter [95% CI 8.12, 18.5]; 91.2 squared T score points per vear of schooling [95% CI 69.1, 113]; both p < 0.001; Fig 3).

Speaking English was strongly positively associated with the Raven's score as well as related to the current geographic location and SRPH. Participants who spoke English were more likely

Table 1. Selected descriptive characteristics of the adolescent cohort.

Characteristic	Total	Male	Female	<i>p</i> -Value
Number of participants	1,463	746 (51.0%); 746 (52.9%)*	717 (49.0%); 663 (47.1%)*	0.3 ^a
Recorded age (years)	22.6 (3.5)	22.5 (3.4)	22.6 (3.6)	0.8 ^b
Height (cm)	162.9 (9.5)	169.5 (7.0)	156.0 (6.4)	<0.001 ^b
Weight (kg)*	57.6 (9.8)	61.5 (9.5)	52.9 (8.1)*	<0.001 ^b
Waist girth (cm)*	78.2 (9.6)	79.8 (8.5)	77.3 (10.1)*	<0.001 ^b
BMI (kg/m ²)*	21.7 (3.3)	21.4 (3.1)	21.8 (3.3)*	0.02 ^b
BMI category*				
Underweight (BMI < 18.5 kg/m ²)	198 (13.5%)	107 (14.3%)	89 (13.4%)*	0.3 ^c
Normal BMI (BMI 18.5–24.9 kg/m ²)	1,065 (72.8%)	555 (74.4%)	477 (71.9%)*	
Overweight (BMI 25.0–29.9 kg/m ²)	171 (11.7%)	74 (9.9%)	83 (12.5%)*	
Obese (BMI \geq 30 kg/m ²)	29 (2.0%)	10 (1.3%)	14 (2.1%)*	
Systolic BP (mm Hg)	113.2 (10.2)	115.2 (9.5)	111.1 (10.4)	<0.001 ^b
Diastolic BP (mm Hg)	75.8 (8.5)	77.0 (8.4)	74.8 (8.4)	<0.001 ^b
BP category**				
Normal BP	1,234 (84.4%)	605 (81.1%)	629 (87.8%)	0.002 ^c
Elevated BP	42 (2.9%)	26 (3.5%)	16 (2.2%)	
Stage 1 hypertension	157 (10.7%)	93 (12.5%)	64 (8.9%)	
Stage 2 hypertension	29 (2.0%)	22 (2.9%)	7 (1.0%)	
Married	211 (14.4%)	41 (5.5%)	170 (23.7%)	< 0.001 ^a
Raven's score (T score)	50.0 (10.0)	51.3 (10.0)	48.6 (9.8)	<0.001 ^b
Currently in Oshikhandass	945 (64.6%)	428 (57.4%)	517 (72.1%)	< 0.001 ^a
Student***	984 (67.3%)	515 (69.0%)	469 (65.4%)	0.2 ^a
Employed***	269 (18.4%)	223 (29.9%)	73 (10.2%)	<0.001 ^a
Highest level of education (years)	11.1 (2.8)	11.0 (2.7)	11.2 (2.9)	0.3 ^b
Speaks English	390 (26.7%)	242 (32.4%)	148 (20.6%)	< 0.001 ^a
Speaks Burushaski	1,010 (69.0%)	528 (70.8%)	482 (67.2%)	0.2 ^a
Speaks Shina	1,414 (96.7%)	729 (97.7%)	685 (95.5%)	0.03 ^a
Mother illiterate	1,031 (70.5%)	524 (70.5%)	507 (71.2%)	>0.999 ^a
Father illiterate	440 (30.1%)	236 (31.9%)	204 (28.8%)	0.75 ^a
CH				<0.001 ^c
Excellent	138 (9.4%)	69 (9.2%)	69 (9.6%)	
Good	294 (20.1%)	153 (20.5%)	141 (19.7%)	
Satisfactory	766 (52.4%)	427 (57.2%)	339 (47.3%)	
Poor	262 (17.9%)	96 (12.9%)	166 (23.2%)	
Very poor	3 (0.2%)	1 (0.1%)	2 (0.3%)	
SRPH				0.09 ^c
Excellent	117 (8.0%)	57 (7.6%)	60 (8.4%)	
Good	388 (26.5%)	194 (26.0%)	194 (27.1%)	
Satisfactory	561 (38.3%)	308 (41.3%)	253 (35.3%)	
Poor	386 (26.4%)	184 (24.7%)	202 (28.2%)	
Very poor	11 (0.8%)	3 (0.4%)	8 (1.1%)	

Data are mean (SD) or *n* (%). BMI, body mass index; BP, blood pressure; SRCH, self-reported current health; SRPH, self-reported past childhood health. *Fifty-four females were pregnant at the time of the interview. These individuals are not included in the weight, waist girth, and BMI measurements.

**Normal except if elevated (systolic blood pressure 121–130 mm Hg and diastolic blood pressure ≤ 80 mm Hg), stage 1 hypertension (systolic blood pressure 131–140 mm Hg or diastolic blood pressure 81–90 mm Hg), or stage 2 hypertension (systolic blood pressure > 140 mm Hg or diastolic blood pressure > 90 mm Hg).

***Categories are not mutually exclusive.

^aTest of proportions.

^b*t*-test for normally distributed data.

^cChi-squared test for categorical variables.

https://doi.org/10.1371/journal.pmed.1003745.t001

to have left Oshikhandass (63% [244/390] had moved away compared to 26% [274/1,071] of non-English speakers, p < 0.001, test of proportions) and to report excellent or good childhood health (39% [154/390] of those who spoke English versus 26% [278/1,071] of non-English speakers, p < 0.001, test of proportions). English was also related to participants' primary language: Burushaski speakers (35%, 286/812) were more likely to speak English than Shina speakers (8%, 29/344, p < 0.001, test of proportions). Speaking English was the only variable with robust evidence to link the childhood home environment (including parental and household characteristics, mediated through maternal education and primary language) and Raven's score. Adolescent participants who spoke English were more likely to have educated parents (41.1% versus 24.8%, p = 0.024, chi-squared test) and to have been raised in higher income households (37.0% versus 21.6%, p = 0.034, chi-squared test) and households with lower household density (2.6 versus 2.7 people/room, p = 0.013, Kruskal–Wallis test) (see S3 Table).

No relationships were identified in the BN for the number of episodes of childhood diarrhea, and no evidence emerged that either diarrhea or pneumonia was related to the major outcomes in separate regression models (see S4 and S5 Tables). In separate regressions, unemployed individuals tended to have significantly higher rates of childhood diarrhea than those employed (see S4 Table), but this association was not supported in the bootstrapped BN.

Self-reported health status

Most participants reported satisfactory, good, or excellent health for both SRPH (72.8%, 849/ 1,165) and SRCH (81.9%, 957/1,165). However, females were more likely than males to report poor/very poor SRCH; this difference was less pronounced in the SRPH (Table 1). The single predictor of SRCH was SRPH (Fig 2). The few individuals who reported poor/very poor SRPH were also most likely to report poor SRCH (Fig 4).

Childhood health status was itself a predictor of lifetime health problems (Fig 2), with participants who characterized their childhood health as poor having an odds ratio of reporting later health problems of 3.4 (95% CI 2.3, 5.3, p < 0.001, multinomial regression; Fig 4). Lower SRPH (poor/satisfactory) was associated with repeated years of schooling.

BMI

Height, weight, and waist girth were significantly higher among males than non-pregnant females, but BMI was slightly higher among females (<u>Table 1</u>). Additionally, the BN showed weak support (50% of bootstrap samples) for English-speaking individuals having a higher BMI (Fig 2).

Education and employment

Participants had 11.1 years of education on average, and 984 (67.3%) were still studying. A higher proportion of males were employed (Table 1). Positive associations existed between years of education and speaking English and between education and employment (Fig 2). During childhood, most adolescents' mothers (1,031, 70.5%), and one-third (440, 30.1%) of fathers were illiterate (Table 1).

Most students (834, 85%) were full-time; a minority (88, 8.9%) were also employed. Of 479 non-students, 283 had completed studies, of whom 208 (73%) were employed, and the remaining 196 (85% of whom, or 166, were female) were doing unpaid family work. Employed participants (n = 313) described a variety of occupations, including teaching (42, 13.4%), business (35, 11.1%), and the army (25, 8.0%), but 106 (33.8%) had higher income roles (e.g., engineering, managerial/financial work, and hotel-related work). In contrast, from the childhood



Fig 2. Bayesian network showing associations that were robust in \geq **50% of 7,500 bootstrap samples.** Continuous nodes are shown as rectangles, and categorical nodes as ellipses; childhood nodes are indicated by dashed outlines, and adolescent nodes by solid outlines. Arc width is proportional to the change in the Bayesian information criterion (BIC): Thicker arrows indicate a larger change in BIC and evidence of a more informative association. Numbers indicate the proportion of bootstrap iterations that supported a given arc. The main outcomes are highlighted (hypertension, Raven's score, SRCH, BMI, educational level, and employment status). The direction of associations is based on the maximum likelihood. BP, blood pressure; HAZ, length/height-for-age *z* score; No., number of; SRCH, self-reported current health; SRPH, self-reported past childhood health.

https://doi.org/10.1371/journal.pmed.1003745.g002

cohort study, their parents described more limited roles for mothers (724/806, 89.8%, were housewives) and for fathers (farming/labor, 35%; the army, 16%; and business, 16%).

Discussion

Human capital is the capability of individuals to contribute to society, and depends, at least in part, on their health, education, employability, and cultural experiences. Adolescents are underrepresented in health research despite their unique role in representing the culmination of childhood life experiences and potentially predicting chronic illness in later life; most relevant literature has focused on the long-term consequences of early childhood malnutrition (see S6 Table). Adolescence also crucially captures the transition from child to actively



Fig 3. Direct predictors of adolescent Raven's T score based on the structure identified in the Bayesian network. The Raven's score is a function of the participants' age, current height, their attained level of education, and whether they spoke English. The light grey 95% confidence intervals indicate the full range of each predictor, and the darker shading indicates the 95% confidence interval across the interquartile range of each respective predictor.

https://doi.org/10.1371/journal.pmed.1003745.g003

contributing adult, including becoming a parent. This study provides a unique opportunity to examine the linkages between childhood and adolescent health status. We hypothesized that the participants' early childhood experiences would affect the development and formation of human capital in this remote Pakistani village.



Fig 4. The mean probability of the level of self-reported current health (SRCH) as an adolescent as a function of self-reported past childhood health (SRPH). Whiskers indicate the 95% confidence intervals. Results based on the Bayesian network. The "very poor" category was pooled with the "poor" category due to the small number of responses.

https://doi.org/10.1371/journal.pmed.1003745.g004

Underpinning individual human capital potential is nonverbal reasoning. We found that, instead of childhood illness (i.e., diarrhea and pneumonia), nonverbal reasoning was more closely related to a collection of variables that shared an association with language, opportunities for education, and the home environment (e.g., maternal education and parental income). Additionally, we found some evidence to suggest that childhood malnutrition (indicated by HAZ) had a small indirect association with nonverbal reasoning. Two meta-analyses by Sudfeld et al. [18] and Victora et al. [19] reported positive associations between HAZ and cognitive scores, using cross-sectional and longitudinal data, respectively. Like the studies reanalyzed by Victora et al. [19], we found a positive association between childhood HAZ and later height; however, we did not find evidence of an association between undernutrition and cognitive development. Our results are consistent with findings that children who recover from stunting show no difference in cognitive outcomes from those who were not stunted [20-22]. This population underwent substantial secular changes, illustrated by the region's increases in per capita GDP, diversification of employment opportunities, and access to education. These factors may have alleviated some of the common constraints on both growth and cognitive development during an important period of development, but are also suggestive of opportunities to intervene that extend beyond early childhood [23].

In earlier analyses of a subset of this cohort (n = 107) [24], childhood morbidity was significantly associated with lower educational performance at 7 years old, consistent with the findings of other studies [25,26]. Education is a major component of the HCI. In 2019, the World Bank estimated that a Pakistani child born in 2018 can expect to complete 8.8 years of school, but only 4.8 years when adjusting for the quality of education [1]. In our population, the mean number of years of education was higher for both females and males, at 11.1, and was significantly higher than in the previous generation, where almost three-quarters of mothers and a third of fathers were illiterate or had no formal schooling. Although it is not possible to determine with these data, school performance, rather than attendance, may still be negatively impacted by childhood malnutrition and morbidity.

Childhood mortality in Oshikhandass decreased significantly from 1989 to 1996 [9], allowing families to focus more on helping surviving children grow and thrive. During this time, educational opportunities significantly improved in Oshikhandass, from there being 1 government high school for boys and 1 middle school for girls to the current situation of 9 early childhood education centers, 11 schools, and 2 colleges (up to class 12). One reason that the Burushaski speakers tended to have more years of education is that the leaders of the Ismaili community (primarily Burushaski speakers in Oshikhandass) emphasized its importance, especially for girls [7]. In Oshikhandass, there were major investments in education, especially improving opportunities for females. Notably, many adolescents in this cohort continued their education at university in Gilgit or elsewhere. Given the importance of this age in developing resilience, understanding how experiences and stresses influence development into adulthood is of great importance [27].

Our key hypothesis was that long-lasting associations between childhood health experiences and adolescent nonverbal reasoning and health status would be identified. We found no evidence for direct associations between the childhood and family factors measured and adolescent nonverbal reasoning. However, participants' mean childhood HAZ, which is an indicator of longer-term nutritional status, was positively correlated with adolescent height (approximately 1 cm per 1 *z*-score change, approximately a third of the association found in other studies [19]) and thereby indirectly to Raven's score (3 T score points per 1 cm adolescent height); this is similar to the findings of the other studies described above, but we did not find the strong associations others did. Our finding that neither childhood diarrhea nor pneumonia was associated with adolescent Raven's score contrasts to studies of cognitive scores in early childhood that report negative impacts of nutrition, infection, and illness [28], but is consistent with findings that such early effects diminish by age 5 years [29]. Given the temporal gap between childhood illness, potential recovery from stunting, and the age at Raven's test administration, this is perhaps not surprising [30].

Previous studies have examined how the mesoenvironment [31], including a nurturing and stimulating home environment [10,32], becomes increasingly influential in child development as a child ages. Whether or not English was spoken by the participant was identified as a variable that linked both to Raven's score and to a suite of variables (parental education and occupation, paternal income, dwelling type, household density, type of toilet, and primary language) that characterized a participant's childhood home environment. We interpreted speaking English as an indication of higher childhood socioeconomic status and greater educational opportunity, as participants from lower socioeconomic backgrounds were less likely to speak English; speaking English was also associated with participants' opportunities to travel beyond Oshikhandass for either work or education.

The Raven's matrices are not typically thought to be biased by educational background and are designed to test observation, clarity of thinking, and nonverbal reasoning skills [12]. Here, we found that participants with more years of education tended to achieve higher Raven's scores. Other studies have similarly shown a positive correlation between educational age (up to a chronological age of 20 years) and Raven's score that might indicate a familiarity with the type of logic assessed, distinct from familiarity with the test per se [33].

SRCH was related to the participant's SRPH, but not to objective measures of health (e.g., BMI, anthropometry, or blood pressure). SRCH was also not related to socioeconomic variables, as might be expected based on the theory underpinning human capital [34]. However, self-reported health remains a meaningful tool and predictor of mortality [35].

BMI is one objective measurement of health with correlations to many chronic conditions that manifest in adulthood—such as diabetes, cancer, osteoarthritis, and cardiovascular diseases [36]—that influence adult survival and therefore the HCI. Blood pressure and hypertension are also objective measures of health, but here only related to weight; neither of these metrics was associated to childhood health except through indirect association with HAZ (via weight and height). Hypertension is reported to be a significant problem in northern Pakistan [37]. It is potentially related to salted tea consumption, and our finding of high rates of diastolic hypertension suggests that it will be important to monitor the prevalence of noncommunicable diseases [4,38] in the future in this area.

Childhood experiences varied by sociocultural group, indicated by the participant's primary language. The Burushaski subpopulation, as compared to the Shina subpopulation, were more likely speak English, to have moved out of the area, to reach higher educational levels, and to have come from families that had a higher socioeconomic status. Similarly, there were notable sex differences, with males tending to have a greater opportunity for continuing their education than females, as they were less likely to have married or to have children, and more likely to have traveled away from Oshikhandass for further study and employment. Additional targeted research would be required to better understand why adolescent females in Oshikhandass reported poorer perceived health, as reported elsewhere in Pakistan [39], and to understand the status of female empowerment.

Following the same cohort from birth to adolescence/young adulthood gives a rare opportunity to understand links from early life experiences to later outcomes, but it comes with the challenge of loss to follow-up. Given that this study was built upon an earlier study, the variables collected were also constrained by what was measured previously (e.g., no cognitive, blood pressure, micronutrient, dietary diversity, or food insecurity indicators were collected during childhood, nor were more detailed parental variables such as maternal mental health). The age at enrollment and duration of initial follow-up also varied. Using the mean HAZ over available HAZ measurements was an attempt to mitigate the variable ages at observation; however, summarizing the data over the first 5 years of life obscures considerable development that takes place in early childhood. Additionally, as the original participants were followed up later, we used retrospective questions to ask about their perceived childhood health at the same time as their perceived current health status. This may lead to biased responses, especially recalling events over several years, and is acknowledged as a subjective rather than objective assessment of health, although we did include other clinical assessments of general health. Another limitation is measurement of school attendance, but not school performance, as has been used elsewhere [26], including in the HCI.

This study does have several strengths; principally, as a prospective longitudinal study spanning over 20 years, it adds significantly to the body of research examining the effects of childhood health and early life exposures on adolescent health status. Additionally, this study examined not only physiological health, but also reasoning abilities and self-perceptions of health.

In summary, we found that adolescents raised in a remote rural village in northeastern Pakistan had high rates of education—significantly higher than their parents—and that most were still students. Early childhood illness (diarrhea or pneumonia) was unrelated to Raven's score, current self-reported health status, BMI, hypertension, education, or employment. The primary language spoken did link childhood socioeconomic factors, characterized by maternal education, to adolescent outcomes. Sex differences were already apparent in these adolescent outcomes. We present evidence that there are opportunities to improve human capital beyond early childhood, and that with targeted investments such gains can be made even over a single generation.

Supporting information

S1 Fig. The distributions of HAZ from childhood measurements and approximated for adolescents. Childhood measurements (dashed lines) and approximations for adolescents (solid lines). The childhood distribution is the mean value of each child's available measurements. Adolescents who were older than 19 years when their height was measured were assumed to be 19 years old because this is maximum age in the WHO growth reference to calculate the age- and sex-standardized HAZ. The black vertical lines indicate the mean childhood and adolescent HAZs. Of the 54.1% (630/1,025) of participants who were stunted as a child (HAZ < -2 SD from the median), 84.9% had recovered by adolescence (i.e., 535/630 had HAZ > -2), but 8.4% who had not been stunted as children became stunted as adolescents (45/535).

(DOCX)

S1 Table. Selected descriptive characteristics comparing those lost to follow-up and those re-enrolled as adolescents.

(DOCX)

S2 Table. Univariate relationships predicting the squared Raven's T score assuming a linear regression between each variable and the squared Raven's T score. (DOCX)

S3 Table. Characteristics of the cohort according to whether adolescents spoke English. (DOCX)

S4 Table. Regression models of the primary outcomes and diarrheal episodes. Showing coefficients from linear models for Raven's T score, BMI, and blood pressure; a Poisson model for education (log count); and a multinomial for employment status (log odds), all as a function of childhood diarrhea episodes. Variable inclusion was based on the Bayesian network. (DOCX)

S5 Table. Regression models of the primary outcomes and pneumonia episodes. Showing coefficients from linear models for Raven's T score, BMI, and blood pressure; a Poisson model for education (log count); and a multinomial for employment status (log odds), all as a function of childhood pneumonia episodes. Variable inclusion was based on the Bayesian network. (DOCX)

S6 Table. Literature search. Forty-five studies identified using 2 search strategies of the PubMed (US National Library of Medicine) database were manually filtered by the first and last author for relevance to this study. Search 1: (cohort OR longitudinal) AND (childhood OR adolescence) AND (cognitive OR cognition) AND (diarrhea OR diarrhoea OR pneumonia OR "childhood illness*") AND (BMI OR "health progression" OR growth OR "child development" OR "childhood development"). Search 1 yielded 21 studies. Search 2: ((child* OR infant OR adolescen*) AND (cohort* OR longitudinal OR followup OR "follow up")) AND cogniti* AND (("human capital" OR "human capacit*" OR potential OR "child development" OR "human capacit*" OR potential OR "child development" OR "halth progression" OR growth OR "child development"). Search 2 yielded 34 studies. And 30 more studies were identified based on reviewers' comments.

(DOCX)

S1 Appendix. STROBE checklist for observational studies. (DOCX)

S2 Appendix. Protocol (adolescent study). (DOCX)

S3 Appendix. Questionnaire (adolescent study). (PDF)

S4 Appendix. Description of the psychometric analysis of the Raven's Standard Progressive Matrices and Colored Progressive Matrices. (DOCX)

S5 Appendix. Protocol and scoring form for Raven's Standard Progressive Matrices and Colored Progressive Matrices administration. (DOCX)

Acknowledgments

We are deeply indebted to the adolescents and their families who participated in this study and the dedicated project staff, including from Aga Khan University: Ahmed Jan and Saba Wasim; Fogarty International Center: Elizabeth Thomas, Faran Sikandar, Julia Baker, and Stacey Knobler; data collectors: Assis Jahan, Faheemullah Beg, Farah Naz Hashmani, Liaqat Ali, Rizwan Karim, and Shafiqa Yar Baig; Karakoram International University: Arif Hussain, Mirza Jibran, and Asif Hussain; study workers: Zohra Bano, Mobina Bano, Gul Nasreen, Mehtab Bano, Kaniz Fatima, Iqbal Bano, Dil Roz, Nazara, Ghazala, Nasima Begum, Alia Rani, Mehwish Hakeem, Rubina, Sameena, Zevar Jan, Sunaira, and Resham Jan. We thank Alicia Livinski, National Institutes of Health Library, for literature searching, editing, and manuscript preparation assistance, and Professors Laura Murray-Kolb and Zulfiqar Bhutta and Drs. David J. Spiro, Peter Kilmarx, Cecile Viboud, and Roger Glass for editorial comments.

The findings and conclusions in this paper are those of the authors and do not necessarily represent the official position of the National Institutes of Health or US Department of Health and Human Services.

Author Contributions

- **Conceptualization:** Zeba A. Rasmussen, Chelsea L. Hansen, Syed Iqbal Azam, Khalil Ahmed, Benjamin J. J. McCormick.
- **Data curation:** Zeba A. Rasmussen, Wasiat H. Shah, Chelsea L. Hansen, Syed Iqbal Azam, Ejaz Hussain, Barbara A. Schaefer, Alexandra F. Jamison, Benjamin J. J. McCormick.
- Formal analysis: Zeba A. Rasmussen, Wasiat H. Shah, Chelsea L. Hansen, Syed Iqbal Azam, Barbara A. Schaefer, Nicole Zhong, Alexandra F. Jamison, Benjamin J. J. McCormick.
- Funding acquisition: Zeba A. Rasmussen, Khalil Ahmed.
- Investigation: Zeba A. Rasmussen, Wasiat H. Shah, Syed Iqbal Azam, Ejaz Hussain.
- Methodology: Zeba A. Rasmussen, Wasiat H. Shah, Syed Iqbal Azam, Ejaz Hussain, Barbara A. Schaefer, Khalil Ahmed.
- **Project administration:** Zeba A. Rasmussen, Wasiat H. Shah, Chelsea L. Hansen, Syed Iqbal Azam, Ejaz Hussain, Khalil Ahmed.
- Resources: Zeba A. Rasmussen.
- Supervision: Zeba A. Rasmussen, Wasiat H. Shah, Syed Iqbal Azam, Ejaz Hussain, Barbara A. Schaefer, Khalil Ahmed.
- Validation: Zeba A. Rasmussen, Wasiat H. Shah, Syed Iqbal Azam, Ejaz Hussain.
- Visualization: Zeba A. Rasmussen, Chelsea L. Hansen, Barbara A. Schaefer, Alexandra F. Jamison, Benjamin J. J. McCormick.
- Writing original draft: Zeba A. Rasmussen, Wasiat H. Shah, Chelsea L. Hansen, Benjamin J. J. McCormick.
- Writing review & editing: Zeba A. Rasmussen, Wasiat H. Shah, Chelsea L. Hansen, Syed Iqbal Azam, Ejaz Hussain, Barbara A. Schaefer, Nicole Zhong, Alexandra F. Jamison, Khalil Ahmed, Benjamin J. J. McCormick.

References

- 1. World Bank. Human capital project. Washington (DC): World Bank; 2019 [cited 2019 Dec 6]. Available from: https://www.worldbank.org/en/publication/human-capital#firstLink01633.
- World Bank. World development report 2019: the changing nature of work. Washington (DC): World Bank; 2019 [cited 2021 May 14]. Available from: https://www.worldbank.org/en/publication/wdr2019.
- Viner RM, Allen NB, Patton GC, Bundy DAP, de Silva N, Horton S, et al. Puberty, developmental processes, and health interventions. In: Bundy DAP, de Silva N, Horton S, Jamison DT, Patton GC, editors. Child and adolescent health and development. 3rd edition. Washington (DC): International Bank for Reconstruction and Development; 2017.
- Sawyer SM, Afifi RA, Bearinger LH, Blakemore SJ, Dick B, Ezeh AC, et al. Adolescence: a foundation for future health. Lancet. 2012; 379(9826):1630–40. https://doi.org/10.1016/S0140-6736(12)60072-5 PMID: 22538178

- 5. World Bank. The Aga Khan Rural Support Program in Pakistan: a second interim evaluation. Washington (DC): World Bank; 1990 [cited 2020 Dec 9]. Available from: <u>https://www.ircwash.org/sites/default/files/Worldbank-1990-AgaKhan.pdf</u>.
- Walraven G, Manaseki-Holland S, Hussain A, Tomaro JB. Improving maternal and child health in difficult environments: the case for "cross-border" health care. PLoS Med. 2009; 6(1):e5. <u>https://doi.org/10. 1371/journal.pmed.1000005 PMID: 19143468</u>
- Benz A. Education and development in the Karakorum: educational expansion and its impacts in Gilgit-Baltistan, Pakistan. Erdkunde. 2013; 67(2):123–36.
- World Bank, Government of Pakistan, Asian Development Bank. Pakistan: Gilgit-Baltistan economic report—broadening the transformation. Islamabad: Government of Pakistan; 2010 [cited 2021 May 18]. Available from: https://portal.pnd.gog.pk/Content/Files/Reports/WB%20Economic%20Report-GB %20pdf_200512092.pdf.
- Hansen CL, McCormick BJJ, Azam SI, Ahmed K, Baker JM, Hussain E, et al. Substantial and sustained reduction in under-5 mortality, diarrhea, and pneumonia in Oshikhandass, Pakistan: evidence from two longitudinal cohort studies 15 years apart. BMC Public Health. 2020; 20(1):759. https://doi.org/10.1186/ s12889-020-08847-7 PMID: 32448276
- World Health Organization, United Nations Children's Fund, World Bank Group. Nurturing care for early childhood development: a framework for helping children survive and thrive to transform health and human potential. Geneva: World Health Organization; 2018 [cited 2021 Jul 13]. Available from: https:// apps.who.int/iris/handle/10665/272603.
- von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, Vandenbroucke JP. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. PLoS Med. 2007; 4(10):e296. https://doi.org/10.1371/journal.pmed.0040296 PMID: 17941714
- 12. Raven J, Raven JC, Court JH. Manual for Raven's progressive matrices and vocabulary scales. Section 1. General overview. Oxford: Oxford Psychologists Press; 1998.
- Murray-Kolb LE, Rasmussen ZA, Scharf RJ, Rasheed MA, Svensen E, Seidman JC, et al. The MAL-ED cohort study: methods and lessons learned when assessing early child development and caregiving mediators in infants and young children in 8 low- and middle-income countries. Clin Infect Dis. 2014; 59 (Suppl 4):S261–72. https://doi.org/10.1093/cid/ciu437
- American Educational Research Association, American Psychological Association, National Council on Measurement in Education. Standards for educational and psychological testing. Washington (DC): American Educational Research Association; 2014.
- Scutari M. Learning Bayesian networks with the bnlearn R package. J Stat Softw. 2010; 35(3):1–22. https://doi.org/10.18637/jss.v035.i03
- Lewis FI, Ward MP. Improving epidemiologic data analyses through multivariate regression modelling. Emerg Themes Epidemiol. 2013; 10(1):4. https://doi.org/10.1186/1742-7622-10-4 PMID: 23683753
- Glover F, Laguna M. Modern heuristic techniques for combinatorial problems. New York: John Wiley & Sons; 1993.
- Sudfeld CR, McCoy DC, Danaei G, Fink G, Ezzati M, Andrews KG, et al. Linear growth and child development in low- and middle-income countries: a meta-analysis. Pediatrics. 2015; 135(5):e1266–75. https://doi.org/10.1542/peds.2014-3111 PMID: 25847806
- Victora CG, Adair L, Fall C, Hallal PC, Martorell R, Richter L, et al. Maternal and child undernutrition: consequences for adult health and human capital. Lancet. 2008; 371(9609):340–57. <u>https://doi.org/10.1016/S0140-6736(07)61692-4 PMID</u>: 18206223
- Alam MA, Richard SA, Fahim SM, Mahfuz M, Nahar B, Das S, et al. Impact of early-onset persistent stunting on cognitive development at 5 years of age: results from a multi-country cohort study. PLoS ONE. 2020; 15(1):e0227839. https://doi.org/10.1371/journal.pone.0227839 PMID: 31978156
- Alam MA, Richard SA, Fahim SM, Mahfuz M, Nahar B, Das S, et al. Correction: Impact of early-onset persistent stunting on cognitive development at 5 years of age: results from a multi-country cohort study. PLoS ONE. 2020; 15(2):e0229663. https://doi.org/10.1371/journal.pone.0229663 PMID: 32078654
- Crookston BT, Penny ME, Alder SC, Dickerson TT, Merrill RM, Stanford JB, et al. Children who recover from early stunting and children who are not stunted demonstrate similar levels of cognition. J Nutr. 2010; 140(11):1996–2001. https://doi.org/10.3945/jn.109.118927 PMID: 20844188
- Black MM, Walker SP, Fernald LCH, Andersen CT, DiGirolamo AM, Lu C, et al. Early childhood development coming of age: science through the life course. Lancet. 2017; 389(10064):77–90. https://doi. org/10.1016/S0140-6736(16)31389-7 PMID: 27717614

- 24. Mitchell J. The long-term association of early childhood diarrhea with school success: a case study from Pakistan. J Educ Int Dev. 2006; 2(2):11. https://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1. 571.5769&rep=rep1&type=pdf
- Dewey KG, Begum K. Long-term consequences of stunting in early life. Matern Child Nutr. 2011; 7 (Suppl 3):5–18. https://doi.org/10.1111/j.1740-8709.2011.00349.x PMID: 21929633
- 26. Fink G, Rockers PC. Childhood growth, schooling, and cognitive development: further evidence from the Young Lives study. Am J Clin Nutr. 2014; 100(1):182–8. <u>https://doi.org/10.3945/ajcn.113.080960</u> PMID: 24808488
- 27. Sahar N, Muzaffar N. role of family system, positive emotions and resilience in social adjustment among Pakistani adolescents. J Educ Health Community Psychol. 2017; 6(2):46–58.
- MAL-ED Network Investigators. Early childhood cognitive development is affected by interactions among illness, diet, enteropathogens and the home environment: findings from the MAL-ED birth cohort study. BMJ Glob Health. 2018; 3(4):e000752. <u>https://doi.org/10.1136/bmjgh-2018-000752</u> PMID: 30058645
- McCormick BJJ, Richard SA, Caulfield LE, Pendergast LL, Seidman JC, Koshy B, et al. Early life child micronutrient status, maternal reasoning, and a nurturing household environment have persistent influences on child cognitive development at age 5 years: results from MAL-ED. J Nutr. 2019; 149(8):1460– 9. https://doi.org/10.1093/jn/nxz055 PMID: 31162601
- Undurraga EA, Behrman JR, Emmett SD, Kidd C, Leonard WR, Piantadosi ST, et al. Child stunting is associated with weaker human capital among native Amazonians. Am J Hum Biol. 2018; 30(1):e23059. https://doi.org/10.1002/ajhb.23059 PMID: 28901592
- **31.** Bronfenbrenner U. The ecology of human development. Cambridge: Harvard University Press; 1979. https://doi.org/10.1016/0006-8993(79)90419-0 PMID: 83896
- Britto PR, Lye SJ, Proulx K, Yousafzai AK, Matthews SG, Vaivada T, et al. Nurturing care: promoting early childhood development. Lancet. 2017; 389(10064):91–102. https://doi.org/10.1016/S0140-6736 (16)31390-3 PMID: 27717615
- Brouwers SA, Van de Vijver FJR, Van Hemert DA. Variation in Raven's Progressive Matrices scores across time and place. Learning Individ Differences. 2009; 19(3):330–8. https://doi.org/10.1016/j.lindif. 2008.10.006
- Borgonovi F, Pokropek A. Education and self-reported health: evidence from 23 countries on the role of years of schooling, cognitive skills and social capital. PLoS ONE. 2016; 11(2):e0149716. <u>https://doi.org/ 10.1371/journal.pone.0149716</u> PMID: 26901130
- 35. Miilunpalo S, Vuori I, Oja P, Pasanen M, Urponen H. Self-rated health status as a health measure: the predictive value of self-reported health status on the use of physician services and on mortality in the working-age population. J Clin Epidemiol. 1997; 50(5):517–28. <u>https://doi.org/10.1016/s0895-4356(97)</u> 00045-0 PMID: 9180644
- National Heart, Lung and Blood Institute. Managing overweight and obesity in adults: systematic evidence review from the obesity expert panel, 2013. Bethesda (MD): National Heart, Lung and Blood Institute; 2013 [cited 2021 May 18]. Available from: https://www.nhlbi.nih.gov/sites/default/files/media/docs/obesity-evidence-review.pdf.
- Shah SM, Luby S, Rahbar M, Khan AW, McCormick JB. Hypertension and its determinants among adults in high mountain villages of the northern areas of Pakistan. J Hum Hypertens. 2001; 15(2):107– 12. https://doi.org/10.1038/sj.jhh.1001131 PMID: 11317189
- Kane JB, Harris KM, Morgan SP, Guilkey DK. Pathways of health and human capital from adolescence into young adulthood. Soc Forces. 2018; 96(3):949–76. <u>https://doi.org/10.1093/sf/sox079</u> PMID: 30555185
- **39.** Ahmad K, Jafar TH, Chaturvedi N. Self-rated health in Pakistan: results of a national health survey. BMC Public Health. 2005; 5:51. https://doi.org/10.1186/1471-2458-5-51 PMID: 15943882